CST Part II Types List of corrections to the 2000/01 lecture notes November 7, 2000

Corrections are underlined or crossed out, if not otherwise commented upon. Page 35: The last line of Slide 29 should read “S^′(σ_i) = T S(σ_i) (for i = 1, . . . , n) and τ^′ = T (τ).” Page 51, Slide 38: The last line of the slide should read “In (gen), α, Γ is the typing environment (Γ_{tv} ⊎ {α}, Γ_{ta}).”

Page 53: The title of Slide 40 should read “Explicitly versus implicitly typed languages”. Page 59, lines -6 and -5: “is (∅, {x_1 : α}), does not satisfy . . . given the typing environment α, x_1 : α.”